Author's response to reviews

Title: Assessment of Arterial Stiffness Using Pulse Wave Velocity in the First Year Post Kidney Transplant: a Prospective Cohort Study

Authors:

Kelly A Birdwell (kelly.birdwell@vanderbilt.edu)
Gilad Jaffe (giladjaf@buffalo.edu)
Aihua Bian (anna.bian@Vanderbilt.edu)
Pingsheng Wu (pingsheng.wu@Vanderbilt.edu)
Talat A Ikizler (alp.ikizler@Vanderbilt.edu)

Version: 2  Date: 4 February 2015

Author's response to reviews: see over
Dear *BMC Nephrology* Editorial Board,

Thank you for considering our manuscript titled “Assessment of Arterial Stiffness Using Pulse Wave Velocity in the First Year Post Kidney Transplant: a Prospective Cohort Study” authored by Kelly Ann Birdwell, MD, MSCI, Gilad Jaffe, BS, Aihua Bian, MPH, Pingsheng Wu, PhD, MS, and Talat Alp Ikizler, MD for the *BMC Nephrology*.

In this revised submission, we have 1) added line numbers 2) added authors’ contributions. No changes to title or content have been made.

Sincerely,

Kelly Birdwell, MD, MSCI  
Division of Nephrology  
Vanderbilt University Medical Center  
Nashville, TN, USA